Author’s response to reviews

Title: The study design and methodology for the ARCHER Study - Adolescent Rural Cohort Study of Hormones, Health, Education, Environments and Relationships.

Authors:

Katharine Steinbeck Prof (kate.steinbeck@health.nsw.gov.au)
Philip Hazell Prof (Philip.hazell@sswahs.nsw.gov.au)
Robert G Cumming Prof (robert.cumming@sydney.edu.au)
Susan R Skinner A/Prof (rachel.skinner@health.nsw.gov.au)
Rebecca Ivers Prof (rivers@georgeinstitute.org.au)
Robert Booy Prof (robert.booy@health.nsw.gov.au)
Greg Fulcher Prof (gfulcher@med.usyd.edu.au)
David J Handelsman Prof (djh@anzac.edu.au)
Andrew J Martin Prof (andrew.martin@sydney.edu.au)
Geoff Morgan Prof (geoff.morgan@neahs.health.nsw.gov.au)
Jean Starling Dr (jean.starling@sydney.edu.au)
Adrian Bauman Prof (adrianb@health.usyd.edu.au)
Margot L Rawsthorne Dr (margot.rawsthorne@sydney.edu.au)
David Bennett Prof (david.bennett@health.nsw.gov.au)
Chin-Moi Chow Dr (chin-moi.chow@sydney.edu.au)
Mary K Lam Dr (mary.lam@sydney.edu.au)
Patrick Kelly Dr (p.kelly@sydney.edu.au)
Ngiare J Brown Dr (nbrown@uow.edu.au)
Karen T Paxton Mrs (karen.paxton@sydney.edu.au)
Catherine Hawke Dr (catherine.hawke@sydney.edu.au)

Version: 2 Date: 16 August 2012

Author’s response to reviews: see over